Author's response to reviews

Title: Risk factors for fatal candidemia caused by Candida albicans and non-albicans Candida species

Authors:

Ming-Fang Cheng (mfcheng@vghks.gov.tw)
Yun-Liang Yang (yyang@mail.nctu.edu.tw)
Tzy-Jyun Yao (tjyao@nhri.org.tw)
Chin-Yu Lin (lin09_1999@yahoo.com)
Jih-Shin Liu (jihshin@nhri.org.tw)
Ran-Bin Tang (rbtang@vghtpe.gov.tw)
Kwok-Woon Yu (kwyu@vghtpe.gov.tw)
Yu-Hua Fan (yhf@vghtpe.gov.tw)
Kai-Sheng Hsieh (kshsieh@isca.vghks.gov.tw)
Monto Ho (monto@imap.pitt.edu)
Hsiu-Jung Lo (hjlo@nhri.org.tw)

Version: 3 Date: 5 April 2005

Author's response to reviews:

April 5, 2005

To whom it may concern,
Enclosed please find the revised version of manuscript 5727355235859146 entitled "Risk factors for fatal candidemia caused by Candida albicans and non-albicans Candida species". We have also revised the manuscript according to the first two comments of the reviewer 1, Dr. Arnaldo L. Colombo. We have also made the changes of formats as requested.

Best regards,
Sincerely yours,

Hsiu-Jung Lo, Ph.D.
Corresponding author